Author’s response to reviews

Title: The Effect of Manual Therapy and Stabilizing Exercises on Forward Head and Rounded Shoulder Postures: a six-week intervention with a one-month follow-up study

Authors:

Kiana Fatholahnejad (kh_letafat@yahoo.com)
Amir Letafatkar (letafatkaramir@yahoo.com)
Malihe hadadnezhad (m.hadadnezhad@yahoo.com)

Version: 1 Date: 02 May 2018

Author's response to reviews:

Dear reviewers

1. The manuscript is edited to eradicate overlapping parts. The previous published papers their methodologies were used in current manuscript are cited.
2. Trail registration date is mentioned after abstract
3. The written consent form was obtained from participant in figure 2 to publish this image.
4. The name of the organization approved the study is mentioned in method section.
5. The unclear sentence in “Ethics approval and consent to participate” section in the end of manuscript is edited and become clear.
6. Data and materials are not applicable because of the restriction applied by the Ethical Committee of the Kharazmi University. Based on this committee, it is crucial for authors to keep any information about subjects safe and confidential.
7. A clean version of manuscript is submitted.